**Author’s response to reviews**

**Title:** WHAT’S THE UPTAKE? Pragmatic RCTs may be used to estimate uptake, and thereby population impact of interventions, but better reporting of trial recruitment processes is needed.

**Authors:**

Katy Bell (katy.bell@sydney.edu.au)

Amanda McCullough (amccullo@bond.edu.au)

Chris Del Mar (cdelmar@bond.edu.au)

Paul Glasziou (pglaszio@bond.edu.au)

**Version:** 3  **Date:** 15 Nov 2017

**Author’s response to reviews:**

15/11/2017

Dear Catherine Rice

Re: Manuscript BMRM-D-17-00133R2

Thank you for the further opportunity to revise and resubmit our paper titled ‘WHAT’S THE UPTAKE? Pragmatic RCTs may be used to estimate uptake, and thereby population impact of interventions, but better reporting of trial recruitment processes is needed.”

We thank the editors and reviewers for their comments, and have made the requested minor revisions. We hope that you will now find our paper suitable for publication in BMC Medical Research Methodology.

Yours sincerely

Katy Bell,

School of Public Health, University of Sydney